Author’s response to reviews

Title: Therapeutic Effects of Ranibizumab in Patients with Polypoidal Choroidal Vasculopathy

Authors:

Xiaoya Gu (guxiaoya727@163.com)
Xiaobing Yu (yuxiaobing@sina.com)
Hong Dai (dai-hong@x263.net)

Version: 3 Date: 12 Jun 2019

Author’s response to reviews:

Editor Comments:

Manuscript BOPH-D-19-00063R2 “Therapeutic Effects of Ranibizumab in Patients with Polypoidal Choroidal Vasculopathy” by Gu et al. authors have provided an appropriate response to the reviewer's criticism; However, several technical issues need to be addressed before a manuscript will be acceptable for publication.

1. English writing of the manuscript needs to improve further. Response: The English in this newly revised manuscript are improved with the help of American Journal Experts.

2. Figure 1 and 2 are inadequate low-resolution (449x332 px which translates at 300 DPI to 3.8x2.8 cm printing size) authors need to generate the higher resolution (at least 5x higher). Simply scaling existing low resolution images of the figures 1 and 2 to a larger size is not correct. Please use GraphPad prism or any other appropriate software to generate high resolution graphs. Response: Figure 1 and 2 are generated as high resolution graphs with 1000 DPI.

3. Figure 3 and 4 are submitted as a collection of raw data, which is inappropriate for publication and challenging to follow for the readership. There is several commercial and free image editing software including Photoshop, GIMP, Adobe Illustrator even Microsoft PowerPoint that allows performing the simple task of arranging the raw data panels into the coherent figure and adding labels. Please refrain submitting raw data in future.

I am attaching the files I have generated from the author's submitted data that illustrate how the figures 3 and 4 should look like as the figure design reference for the authors. Please submit the revised copy of the manuscript with improved English writing and redesigned figures. Response: Figure 3 and 4 are redesigned and submitted as suggested.